Author’s response to reviews

Title: Offering Mental health first aid to a person with depression: A Delphi study to re-develop the guidelines published in 2008

Authors:

Kathy Bond (kathybond@mhfa.com.au)
Fairlie Cottrill (fairliec@mhfa.com.au)
Fiona Blee (fionab@mhfa.com.au)
Claire Kelly (clairek@mhfa.com.au)
Betty Kitchener (bettyakitchener@gmail.com)
Anthony Jorm (ajorm@unimelb.edu.au)

Version: 1 Date: 03 Oct 2018

Author’s response to reviews:

Point 1 - I have reworded the section highlighted in your letter.

Point 2 - I have included the suggested statement in the declarations section.